Author’s response to reviews

Title: “Whatever Happens, Happens” Challenges of End-of-Life Communication from the Perspective of Older Adults and Family Caregivers: A Qualitative Study

Authors:

Jennifer im (jennifer.im@mail.utoronto.ca)

Susanna Mak (susanna.mak@sinahealthsystem.ca)

Ross Upshur (ross.upshur@gmail.com)

Leah Steinberg (leah.steinberg@sinahealthsystem.ca)

Kerry Kuluski (Kerry.Kuluski@thp.ca)

Version: 3 Date: 19 Nov 2019

Author’s response to reviews:

Dear Editor,

Thank you for reviewing the manuscript. In the last minor revision, it was requested that I amend the quotations in Table 4 by providing ages as age-ranges, or removing the age (or sex) information. I've removed the sex of the participants. The accompanying characteristics of participants in the quotations now only include their age and the number of chronic conditions. The same edits were applied to the in-text quotations and the manuscript has also been re-uploaded.

Thank you,

Jennifer Im